Author's response to reviews

Title: The effect of taping versus semi-rigid bracing on patient outcome and satisfaction in ankle sprains: a prospective, randomized controlled trial

Authors:

Sacha Lardenoye (s.lardenoye@mumc.nl)
Ed Theunissen (e.theunissen@mumc.nl)
Berry Cleffken (bcl@mumc.nl)
Peter RG Brink (p.brink@mumc.nl)
Rob A de Bie (RA.debie@maastrichtuniversity.nl)
Martijn Poeze (m.poeze@mumc.nl)

Version: 2 Date: 9 August 2011

Author's response to reviews:

Dear Ms. Abigail Quiniquini
Journal Editorial Office
BioMed Central

In response to your email we hereby send you the revised manuscript ID 2016726115519559. Below we will address the answers mentioned in your mail point by point:

1. The trial is registered in the Controlled Clinical Trial Registry under number ISRCTN58932628. This has also been added to the abstract.

2. We unintentionally omitted the approval of the METC of our institution, which is revised in the methods section of the revised manuscript.